Author’s response to reviews

Title: Pleural fluid soluble triggering receptor expressed on myeloid cells-1 as a marker of bacterial infection: a meta-analysis

Authors:

Hanssa Summah (hanssa_04@hotmail.com)
Li-Li Tao (ivydoctao@gmail.com)
Ying-Gang Zhu (robinzyg-zhu@vip.sina.com)
Hong-Ni Jiang (jhn2001@sina.com)
Jie-Ming Qu (jmqu64@yahoo.com.cn)

Version: 6 Date: 15 October 2011

Author’s response to reviews: see over
Dear Editor,

Thank you for pointing out our mistake. Indeed there are 6 figures in altogether; Figure 7 has been wrongly cited. We have therefore changed Figure 7 cited in the main text (page 16) to Figure 6.

Kind regards,
Dr Hanssa Summah,
Dr Li-Li Tao,
Dr Ying-Gang Zhu,
Dr Hong-Ni Jiang,
Prof. Jie-ming Qu.